PHYS 410 Fall 2001 Computational Physics Initial Survey

Name:

Name of account (or “none”) on Physics & Astronomy Sun server, physics.ubc.ca:

Other machines/systems which you can access, including home systems: (For home systems indicate what type of Internet access you have.)

Previous and concurrent physics courses:

Previous and concurrent math courses:

Previous and concurrent computer science courses:

Programming language you know best:

Text editors and/or word processors you are comfortable with:

Level of proficiency (use 0-5 for “none” through “fluent”) in:

Unix: C: C++:

Fortran 77/Fortran 90: Java:

Maple:

Mathematica:

Netscape/Explorer/Other Web Browser (specify):